6.045J/18.400J: Automata, Computability and Complexity Nancy Lynch, Nati Srebro

Handout 10: Counter Machines

March 29, 2004 Susan Hohenberger

This handout will be distributed during Lecture 14 of 6.045 on March 29, 2004. It will not appear on the course website, but students needing a copy may obtain one by contacting the TA. Students will be responsible for this material on Quiz 2 and the final exam.

Material in handout:

Section 8.5 - Restricted Turing Machines Pages 345-355 From Introduction to Automata Theory, Languages and Computation By J. Hopcroft, R. Motwani and J. Ullman

10-1